Author's response to reviews

Title: Cowden Syndrome: The key to managing the risk of cancer

Authors:

Hammami Sonia (soniawhammami@yahoo.fr)
Berriche Olfa (olfaberriche@yahoo.fr)
Belhadjali Hichem (Belhadjalihichem@yahoo.fr)
Mahjoub Sylvia (sylviamahjoub@yahoo.fr)

Version: 2 Date: 20 November 2011

Author's response to reviews: see over
Monastir, November 19, 2011

Editor in Chief
Journal of Medical case report
Manuscript Title : Cowden Syndrome: The key to managing the risk of cancer

Dear Editor in Chief,

We have submit revision of our manuscript, according your suggestion and authors instructions, entitled: "Cowden Syndrome: The key to managing the risk of cancer" in your journal. It contains original rare case, with clinical and therapeutic impact, which has not been submitted for publication elsewhere. We hope that this paper is suitable for publication as case report.

We thank you in advance for the consideration given this paper and look forward to a favourable response from you soon.

Best regards
Dr Sonia Hammami
Department of Internal Medicine
CHU F Bourguiba Monastir (Tunisia)